\begin{table}
\caption{\label{logbook}Summary of observations.}
\begin{tabular}{l l l l l l l l}
\hline\hline
QSO           &  $z_{\rm em}$  & $R$ & RA(2000) & Dec(2000) & Obs. Date & Exp. Time (min) &	Note$^*$	\\
\hline
PSS0007+2417  & 4.050 & 18.69 & 00 07 38.7  & +24 17 25.0  & 2000 Sep. 06 & 60 &  \\ 
PSS0014+3032  & 4.470 & 18.81 & 00 14 42.8  & +30 32 03.0  & 2000 Sep. 06 & 60 &  \\
PSS0052+2405  & 4.280 & 18.30 & 00 52 06.8  & +24 05 39.0  & 1999 Dec. 30 & 40 &  \\ 
PSS0117+1552  & 4.244 & 18.60 & 01 17 31.2  & +15 52 16.4  & 2000 Sep. 04 & 50 & 2\\  
PSS0118+0320  & 4.232 & 18.50 & 01 18 52.8  & +03 20 50.0  & 2000 Sep. 06 & 60 & 3\\  
PSS0121+0347  & 4.127 & 17.86 & 01 21 26.2  & +03 47 07.0  & 1999 Dec. 30 & 45 & 3\\
SDSS0127-0045 & 4.084 & 18.37 & 01 27 00.7	& --00 45 59.4  & 2001 Jan. 02 & 60 &  \\    
PSS0131+0633  & 4.430 & 18.24 & 01 31 12.2  & +06 33 40.0  & 2000 Sep. 04 & 60 & 1, 2\\  
PSS0133+0400  & 4.154 & 17.86 & 01 33 40.4  & +04 00 59.0  & 1999 Dec. 29 & 40 & 1, 2, 3\\  
PSS0134+3307  & 4.536 & 18.82 & 01 34 21.6  & +33 07 56.5  & 2000 Sep. 06 & 60 & 1, 2\\ 
PSS0207+0940  & 4.136 & 18.63 & 02 07 03.5  & +09 40 59.0  & 2000 Sep. 04 & 75 &  \\
SDSS0206+1216 & 4.810 & 21.51 & 02 06 51.4  & +12 16 24.4  & 2002 Dec. 07 & 30 &  \\
PSS0209+0517  & 4.194 & 17.36 & 02 09 44.7  & +05 17 14.0  & 1999 Dec. 29 & 35 & 1, 2, 3\\
SDSS0210-0018 & 4.700 & 20.74 & 02 10 43.2  & --00 18 18.5  & 2002 Dec. 07 & 40 & 4, 5\\   
PSS0211+1107  & 3.975 & 18.12 & 02 11 20.1  & +11 07 16.0  & 1999 Dec. 29 & 70 &  \\
SDSS0211-0009 & 4.900 & 22.04 & 02 11 02.7  & --00 09 10.3  & 2002 Dec. 07 & 20 & 1, 2\\
SDSS0231-0728 & 5.410 & 21.54 & 02 31 37.6  & --07 28 54.5  & 2002 Dec. 06 & 90 & 4, 5\\ 
PSS0244-0108  & 3.990 & 19.00 & 02 44 57.2  & --01 08 08.7  & 2000 Sep. 06 & 40 & 4, 5\\
PSS0248+1802  & 4.422 & 18.40 & 02 48 54.3  & +18 02 50.3  & 1999 Dec. 30 & 60 & 1, 2\\
PSS0320+0208  & 3.960 & 18.74 & 03 20 42.7  & +02 08 16.0  & 1999 Dec. 30 & 55 &  \\
SDSS0338+0021 & 5.020 & 21.68 & 03 38 29.3  & +00 21 56.5  & 2002 Dec. 06 & 90 & 1, 2, 4, 5\\
SDSS0338-RD657& 4.960 & 23.00 & 03 38 31.3  & +00 18 07.7  & 2001 Jan. 02 & 120&  \\
PSS0452+0355  & 4.420 & 18.80 & 04 52 51.5  & +03 55 58.0  & 1999 Dec. 30 & 60 &  \\    
PSS0747+4434  & 4.435 & 18.06 & 07 47 50.0  & +44 34 16.0  & 1999 Dec. 30 & 90 & 1, 2, 5\\   
SDSS0756+4104 & 5.090 & 21.70 & 07 56 18.0  & +41 04 10.6  & 2001 Mar. 26 & 150& 4\\ 
PSS0808+5215  & 4.510 & 18.82 & 08 08 49.5  & +52 15 16.0  & 2000 Apr. 28 & 70 & 5\\  
SDSS0810+4603 & 4.074 & 18.67 & 08 10 54.7  & +46 03 55.2  & 2001 Mar. 24 & 60 &  \\
PSS0852+5045  & 4.200 & 19.00 & 08 52 27.4  & +50 45 11.0  & 2000 May 13 & 60 &  \\
PSS0926+3055  & 4.190 & 17.31 & 09 26 36.3  & +30 55 06.0  & 1999 Dec. 29 & 40 &  \\
SDSS0941+5947 & 4.820 & 20.66 & 09 41 08.4  & +59 47 25.8  & 2002 Dec. 06 & 40 & 4, 5\\
BR0945-0411   & 4.130 & 18.80 & 09 47 49.6  & --04 25 15.1  & 2000 May 14 & 60 &  \\   
PSS0950+5801  & 3.973 & 17.38 & 09 50 14.0  & +58 01 38.0  & 1999 Dec. 29 & 40 & 4, 5\\ 
BR0951-0450   & 4.350 & 18.90 & 09 53 55.7  & --05 04 19.5  & 2000 May 15 & 53 & 2\\   
PSS0955+5940  & 4.340 & 17.84 & 09 55 11.3  & +59 40 32.0  & 1999 Dec. 29 & 56 & 4, 5\\  
PSS0957+3308  & 4.283 & 17.59 & 09 57 44.5  & +33 08 23.0  & 1999 Dec. 29 & 55 & 5\\    
BRI1013+0035  & 4.380 & 18.80 & 10 15 49.0  & +00 20 19.0  & 2000 May 15 & 60 & 2\\  
PSS1026+3828  & 4.180 & 18.93 & 10 26 56.7  & +38 28 43.0  & 2000 Apr. 29 & 95 & 5\\ 
BR1033-0327   & 4.509 & 18.50 & 10 36 23.7  & --03 43 20.0  & 2000 May 13 & 20 &  \\
PSS1039+3445  & 4.390 & 19.20 & 10 39 19.3  & +34 45 10.9  & 2001 Apr. 18 & 150& 5\\
SDSS1044-0125 & 5.740 & 25.10 & 10 44 33.0  & --01 25 03.1  & 2000 Apr. 28 \& 2001 Jan. 01,02 & 330 & \\
SDSS1048+4637 & 6.230 & 22.40 & 10 48 45.0  & +46 37 18.3  & 2003 Jun. 02 & 90 & 5\\
PSS1048+4407  & 4.450 & 19.50 & 10 48 46.6  & +44 07 12.7  & 2000 Apr. 29 & 20 &  \\    
PSS1057+4555  & 4.126 & 17.70 & 10 57 56.4  & +45 55 51.9  & 1999 Dec. 30 & 40 & 1, 2, 5\\   
PSS1058+1245  & 4.330 & 18.00 & 10 58 58.5  & +12 45 55.0  & 1999 Dec. 30 & 75 &  \\  
PSS1118+3702  & 4.030 & 18.76 & 11 18 56.2  & +37 02 53.9  & 2001 Mar. 24 & 60 & 5\\ 
PSS1140+6205  & 4.509 & 18.73 & 11 40 09.6  & +62 05 23.3  & 2000 May 14 & 60 & 4, 5\\ 
PSS1159+1337  & 4.081 & 18.50 & 11 59 06.5  & +13 37 37.8  & 2000 Feb. 10 & 55 & 1, 2, 5\\
SDSS1204-0021 & 5.030 & 20.82 & 12 04 41.7  & --00 21 49.6  & 2003 Jun. 04 & 60 & 4, 5\\
SDSS1208+0010 & 5.273 & 22.75 & 12 08 23.9  & +00 10 28.9  & 2001 Mar. 24 & 120&  \\
PSS1226+0950  & 4.340 & 18.78 & 12 26 23.8  & +09 50 03.7  & 2001 Mar. 24 & 80 &  \\
PSS1247+3406  & 4.897 & 20.40 & 12 49 42.2  & +33 49 54.0  & 2001 Mar. 26 & 60 &  \\
PSS1248+3110  & 4.346 & 18.90 & 12 48 20.2  & +31 10 44.0  & 2000 May 13 & 60 &  \\   
PSS1253-0228  & 4.007 & 19.40 & 12 53 36.3  & --02 28 08.0  & 2000 Apr. 29 & 55 & 1,2\\ 
SDSS1310-0055 & 4.151 & 18.85 & 13 10 52.6  & --00 55 31.8  & 2001 Mar. 24 & 50 & 4, 5\\
PSS1315+2924  & 4.180 & 18.48 & 13 15 39.6  & +29 24 39.8  & 2001 Mar. 24 & 85 &  \\ 
PSS1317+3531  & 4.369 & 19.10 & 13 17 43.3  & +35 31 33.1  & 2001 Mar. 26 & 55 & 2\\
J1325+1123    & 4.400 & 18.77 & 13 25 12.6  & +11 23 32.8  & 2001 Apr. 18 & 120& 5\\    
PSS1326+0743  & 4.123 & 17.30 & 13 26 11.9  & +07 43 59.0  & 2000 Dec. 29 & 60 &  \\
PSS1339+5154  & 4.080 & 18.70 & 13 39 13.0  & +51 54 04.0  & 2000 May 15 & 60 & 5\\
PSS1347+4956  & 4.560 & 17.90 & 13 47 43.4  & +49 56 21.0  & 2000 Feb. 10 & 50 & 5\\
PSS1401+4111  & 4.026 & 18.62 & 14 01 32.8  & +41 11 49.9  & 2000 Apr. 28 & 46 & 4, 5\\
PSS1403+4126  & 3.862 & 18.92 & 14 03 55.7  & +41 26 16.2  & 2000 May 15 & 90 & 4, 5\\
PSS1418+4449  & 4.280 & 18.40 & 14 18 31.8  & +44 49 37.0  & 2000 Feb. 29 & 60 & 4, 5\\
PSS1430+2828  & 4.306 & 19.30 & 14 30 31.9  & +28 28 34.1  & 2001 Mar. 26 & 55 & 2\\
PSS1432+3940  & 4.292 & 18.60 & 14 32 24.9  & +39 40 24.0  & 2000 May 13 & 60 & 4, 5\\     
PSS1435+3057  & 4.350 & 19.30 & 14 35 23.3  & +30 57 16.3  & 2001 Apr. 18 & 70 & 2\\   
PSS1443+2724  & 4.406 & 19.30 & 14 43 31.2  & +27 24 37.0  & 2000 Apr. 29 & 30 & 2\\
PSS1443+5856  & 4.270 & 17.80 & 14 43 40.8  & +58 56 53.0  & 2000 Apr. 28 & 40 & 4, 5\\
PSS1458+6813  & 4.291 & 18.67 & 14 58 31.7 	& +68 13 05.2  & 2000 Apr. 28 & 60 &  \\  
PSS1500+5829  & 4.224 & 18.60 & 15 00 07.7  & +58 29 38.0  & 2000 May 14 & 60 &  \\  
PSS1506+5220  & 4.180 & 18.10 & 15 06 54.6  & +52 20 05.0  & 2000 Feb. 10 & 50 & 4, 5\\
GB1508+5714   & 4.304 & 18.90 & 15 10 02.2  & +57 03 04.9  & 2000 May 13 & 75 & 2\\  
PSS1531+4157  & 4.200 & 19.00 & 15 31 29.4  & +45 17 07.9  & 2001 Apr. 18 & 128&  \\  
PSS1535+2943  & 3.972 & 18.90 & 15 35 53.9  & +29 43 13.0  & 2000 May 15 \& 2000 Sep. 05 & 120 & \\
PSS1543+3417  & 4.390 & 18.40 & 15 43 40.4  & +34 17 45.0  & 2000 Apr. 30 & 76 & 5\\                 
PSS1554+1835  & 3.990 & 18.90 & 15 35 53.9  & +29 43 13.0  & 2000 May 15 \& 2000 Sep. 04 & 120 &\\   
PSS1555+2003  & 4.228 & 18.90 & 15 55 02.6  & +20 03 25.0  & 2000 May 14 & 90 &  \\ 
PSS1615+1803  & 4.010 & 18.42 & 16 15 22.9  & +18 03 56.4  & 2000 Apr. 29 & 90 &  \\
SDSS1630+4012 & 6.050 & 20.42 & 16 30 33.9  & +40 12 09.6  & 2003 Jun. 02,03,04 &  295 & \\
PSS1633+1411  & 4.349 & 19.00 & 16 33 19.7  & +14 11 42.6  & 2000 Apr. 30 & 60 & 1, 2\\  
PSS1646+5514  & 4.084 & 18.11 & 16 46 56.3  & +55 14 46.7  & 2000 Apr. 28 & 60 & 1, 2\\
VLA1713+4218  & 4.230 & 19.00 & 17 13 56.2  & +42 18 08.6  & 2001 Apr. 17 & 90 &  \\   
PSS1715+3809  & 4.520 & 18.56 & 17 15 39.5  & +38 09 06.6  & 2000 Sep. 03 & 60 &  \\  
PSS1721+3256  & 4.040 & 19.23 & 17 21 06.7  & +32 56 35.8  & 2000 May 01 & 60 & 1, 2\\
PSS1723+2243  & 4.514 & 18.17 & 17 23 23.1  & +22 43 56.4  & 2001 Mar. 24 & 40 &  \\
SDSS1737+5828 & 4.940 & 20.93 & 17 37 44.0  & +58 28 25.4  & 2001 Mar. 24 & 90 & 4, 5\\  
PSS1745+6846  & 4.130 & 19.12 & 17 45 50.1  & +68 46 21.0  & 2000 May 01 & 120 & \\  
PSS1802+5616  & 4.158 & 19.19 & 18 02 48.9  & +56 16 51.0  & 2000 Sep. 05 & 90 & 1, 2\\   
PSS2122-0014  & 4.114 & 19.13 & 21 22 07.5  & --00 14 45.0  & 2000 Sep. 06 & 60 & 1, 2, 4, 5\\   
PSS2154+0335  & 4.360 & 18.41 & 21 54 06.9  & +03 35 40.0  & 2000 May 14 & 60 & 1, 2, 3\\
PSS2155+1358  & 4.256 & 18.50 & 21 55 02.1  & +13 58 26.0  & 2000 May 15 & 40 & 1, 2, 3 \\   
PSS2203+1824  & 4.375 & 18.74 & 22 03 43.4  & +18 28 14.0  & 2000 Sep. 04 & 60 &  \\  
PSS2238+2603  & 4.031 & 18.85 & 22 38 41.1  & +26 03 46.0  & 2000 Sep. 06 & 60 &  \\  
PSS2241+1352  & 4.441 & 18.69 & 22 41 47.7  & +13 52 05.9  & 2000 Sep. 03 & 60 & 1, 2, 4, 5\\
PSS2244+1005  & 4.040 & 18.92 & 22 44 05.5  & +10 47 38.0  & 2000 Sep. 04 & 80 &  \\  
PSS2315+0921  & 4.412 & 18.96 & 23 15 59.2  & +09 21 44.0  & 2000 Sep. 05 & 90 &  \\  
PSS2322+1944  & 4.170 & 18.29 & 23 22 07.2  & +19 44 23.0  & 2000 May 13 \& 2000 Sep. 04 & 111 & \\
PSS2323+2758  & 4.180 & 18.51 & 23 23 41.0  & +27 58 01.0  & 2000 Sep. 04 & 86 &  \\  
PSS2344+0342  & 4.340 & 17.87 & 23 44 03.2  & +03 42 26.0  & 1999 Dec. 30 & 45 & 1, 2 \\  
\hline 
\end{tabular}
\begin{flushleft}
 $^*$ Quasars from this survey were previously used for $\Omega_{\rm DLA, HI}$ estimations by: (1) P\'eroux et~al. (\cite{eroux2001});  
 (2) P\'eroux et~al. (\cite{eroux2003}); (3) P\'eroux et~al. (\cite{eroux2005}); (4) Prochaska et~al. (\cite{Prochaska2005}); (5) Prochaska \& Wolfe (\cite{Prochaska2009}). 
\end{flushleft}
\end{table}